Author’s response to reviews

Title: LOST to follow-up Information in Trials (LOST-IT): a protocol on the potential impact

Authors:

Elie A Akl (elieakl@buffalo.edu)
Matthias Briel (mbriel@uhbs.ch)
John J You (jyou@mcmaster.ca)
Francois Lamontagne (francois.lamontagne@usherbrooke.ca)
Azim Gangji (gangji@mcmaster.ca)
Tali Cukierman-Yaffe (cukierm@mcmaster.ca)
Mohamed Alshurafa (alshurafa@mcmaster.ca)
Xin Sun (sunx26@mcmaster.ca)
Kara A Nerenberg (nerenbk@mcmaster.ca)
Bradley C Johnston (bjohnston@med.ualberta.ca)
Claudio Vera (cmverapg@med.uc.cl)
Ed Mills (emills@ccnm.edu)
Dirk Bassler (dirk.bassler@med.uni-tuebingen.de)
Arturo Salazar (esalazar@med.wayne.edu)
Neera Bhatnagar (bhatnag@mcmaster.ca)
Jason W Busse (j.busse@utoronto.ca)
Zara Khalid (khalidz@mcmaster.ca)
Stephen D Walter (walter@mcmaster.ca)
Deborah J Cook (debcook@mcmaster.ca)
Holger J Schunemann (hjs@buffalo.edu)
Douglas G Altman (doug.altman@cancer.org.uk)
Gordon H Guyatt (guyatt@mcmaster.ca)

Version: 4 Date: 9 June 2009

Author’s response to reviews:

Editors-in-Chief, Trials

RE: “LOST to follow-up Information in Trials (LOST-IT): a protocol on the potential impact”

Dear Editors,

We are very delighted that Trials will publish our manuscript. We have made all the formatting changes as suggested.

With kind regards,

Gordon Guyatt, MD, MSc
Departments of Medicine and Biostatistics & Clinical Epidemiology